Author's response to reviews

Title: Right Coronary Artery Originating from Left Anterior Descending Artery; Case Report Hilmi Tokmakoglu Tekden Hospital, Cardiovascular Surgery Department, Kocasinan-Kayseri-Turkey. E-mail: h.tokmakoglu@isnet.net.tr
Orhan Bozoglan Tekden Hospital, Cardiovascular Surgery Department, Kocasinan-Kayseri-Turkey. E-mail: orhanbozoglan1975@hotmail.com
Levent Ozdemir Tekden Hospital, Cardiology Department, Kocasinan-Kayseri-Turkey. E-mail: levent.ozdemir@bozok.edu.tr

Corresponding Author Hilmi Tokmakoglu
Tekden Hastanesi Gevher Nesibe Mah. Gur sok. No: 4 Kocasinan- Kayseri-Turkiye

Authors:

Hilmi Tokmakoglu (h.tokmakoglu@isnet.net.tr)
Orhan Bozoglan (orhanbozoglan1975@hotmail.com)
Levent Ozdemir (levent.ozdemir@bozok.edu.tr)

Version: 2 Date: 20 May 2010

Author's response to reviews:

Dear Editor,

Here is the response point by point to the concerns;

Yours sincerely

Referee 4: All references were numbered in square brackets in the manuscript.

Referee 6:
Two new figures is added to the manuscript to show the course of RCA in right atrioventricular groove and its continuation.